Author's response to reviews

Title: Epidural lipomatosis and congenital small spinal canal in spinal anesthesia: a case report

Authors:

Per Flisberg (per.flisberg@skane.se)
Owain Thomas (owain.thomas@skane.se)
Bo Geijer (bo.geijer@lthalland.se)
Ulf Schott (ulf.schott@skane.se)

Version: 8 Date: 24 November 2008

Author's response to reviews: see over
Reviewer Dahlgren really does not give us any criticism so I leave him out. I think the improved English in the manuscript clears up our message as Dahlgren has proposed. Prof Pinkhard comments have been met by discussing the Borre scale and his own criticism of it. We have updated several references about epidural lipomatosis - see changes in the reference list. A paper by Pinkhard has been discussed and given as an example of a recent publication. Borre scale has been involved in the description of our case and in the legends to figures. The language has been extensively corrected and the manuscript has been rewritten and clarified both by content and by the linguistic correction - i.e., argumentation restructured - both in the abstract, in the case presentation, in the discussion and in the conclusion.